Author's response to reviews

**Title:** Atrial fibrillation in healthy adolescents following highly caffeinated beverage consumption: a case series

**Authors:**

Jennifer R Di Rocco (jdirocco@mcw.edu)
Adelaide During (draduring2004@yahoo.com)
Peter J Morelli (peter.morelli@stonybrook.edu)
Marybeth Heyden (marybeth.heyden@stonybrook.edu)
Thomas A Biancaniello (tbiancan@notes.cc.sunysb.edu)

**Version:** 3  **Date:** 31 August 2010

**Author's response to reviews:** see over
August 30, 2010

Journal of Medical Case Reports
Cover Letter of Revision Changes
JMCR MS ID: 9777100873236786

To Whom It May Concern:

I have made the changes you have requested as outlined below, with my response in red.

Formatting changes to be made:
---------------------------------
Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order.

I have edited the list of authors.

Patient ID: Please ensure the patient identification (age, gender and ethnicity) is stated consistently in the manuscript and the abstract.

I have added the patient’s race and specific ages to both the manuscript and the abstract.

References: Please provide a title for the website in reference 11

I have provided the title for the FDA website *FDA Basics: “Why isn't the amount of caffeine a product contains required on a food label?”* for reference 11.

Minor revisions

Track changes
I have continued to track changes throughout the revision process. If this is not what you meant here, please inform me.

Corresponding author: one or more corresponding author(s) must be indicated on the title page, and their email address must be given
I highlighted Peter J Morelli as our corresponding author, indicated by an asterisk, as outlined in your guide.

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submissions system
Qualifications have been removed.

Affiliations: Please remember to include the country in the affiliation details
I included “USA” in all the affiliation details.

Structure: Please ensure the Competing interests section precedes the Consent section. I switched the order of these sections to comply.

Figure 5: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

I removed the “Figure 5” title from the top of the figure, and cropped the white space as closely as possible.

Figure legends: Please provide the title ‘Figure legends’ for this section

I provided this title.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

I did a final spelling and grammar check.

Thank you for your time. I appreciate the opportunity to publish in your journal.

Sincerely,

Jennifer R. Di Rocco